Author's response to reviews

Title: What makes people decide who to turn to when faced with a mental health problem? Results from a French survey

Authors:

Viviane Kovess-Masfety (vkovess@mgen.fr)
Delphine Saragoussi (delsara@club-internet.fr)
Christine Sevilla-Dedieu (csevilla@mgen.fr)
Fabien Gilbert (fgilbert@mgen.fr)
Agnieszka Suchocka (agnieszka.capuano@curie.net)
Nathalie Arveiller (nathaliearveiller@yahoo.fr)
Isabelle Gasquet (isabelle.gasquet@sap.aphp.fr)
Nadia Younes (NYOUNES@ch-versailles.fr)
Marie-Christine Hardy-Bayle (MCHardyBayle@ch-versailles.fr)

Version: 7 Date: 27 July 2007

Author's response to reviews: see over
27 July 2007

*BMC Public Health*

Manuscript #1403400328127927 – What makes people decide who to turn to when faced with a mental health problem? Results from a French survey

Dear Editor,

Please find attached the revised version of our manuscript.

As requested, the width of the figure has been decreased.

The manuscript has been copy-edited.

Yours sincerely,

Viviane Kovess-Masfety